Author's response to reviews

Title: Evaluation of the effects of Botulinum toxin A injections when used to improve ease of care and comfort in children with cerebral palsy whom are non-ambulant: A double blind randomized controlled trial.

Authors:

Megan Thorley (m.thorley@uq.edu.au)
Samantha Donaghey (Samantha_Donaghey@health.qld.gov.au)
Priya Edwards (Priya_Edwards@health.qld.gov.au)
Lisa Copeland (Lisa_Copeland@health.qld.gov.au)
Megan Kentish (Megan_Kentish@health.qld.gov.au)
Kim McLennan (Kim_McLennan@health.qld.gov.au)
Jayne Lindsley (Jayne_Lindsley@health.qld.gov.au)
Laura Gascoigne-Pees (Laura_Gascoignepees@health.qld.gov.au)
Leanne Sakzewski (leanne.sakzewski@uqconnect.edu.au)
Roslyn Boyd (r.boyd@uq.edu.au)

Version: 2 Date: 2 July 2012

Author's response to reviews: see over